Author's response to reviews

Title: Comprehensive Traditional Chinese Medical Rehabilitation for the Management of Knee Osteoarthritis in Community Health Centre: Study Protocol for a Randomized Controlled Trial

Authors:

Yan Hu (abcdyh@hotmail.com)
Su Y Xin (suyouxin777@hotmail.com)
Chen L Dian (lidianchen87@hotmail.com)
Zheng G Hua (zhgh1969@yahoo.com.cn)
Lin X Ti (linxueyi7777@163.com)
Chen B Jun (chenbaojun133@163.com)
Zhou B Hong (zhoubihong06320@163.com)
Zhang Qing (zhangqing8607@163.com)

Version: 2 Date: 6 September 2013

Author’s response to reviews:

Answer to reviewer:

“Please have the manuscript edited to correct errors in English spelling and grammar. I will review the manuscript in detail once editing is complete.”

Answer: We have substantially revised our manuscript after reading the comments provided and have employed an English-language editing service, LetPub, to polish our wording. Certification is attached.